Author’s response to reviews

Title: Does being physically active prevent future disability in older people? Attenuated effects when taking time-dependent confounders into account

Authors:

Stefan Kreisel (Stefan.Kreisel@evkb.de)
Christian Blahak (christian.blahak@umm.de)
Hansjörg Bätzner (h.baezner@klinikum-stuttgart.de)
Michael Hennerici (michael.hennerici@umm.de)

Version: 2 Date: 23 Oct 2017

Author’s response to reviews:

Dear editors - thank you for letting us revise the manuscript.

We have

- Removed the previous cover letter
- Included a "Consent for publication" section under "Declarations"
- Abbreviated the authors in the "Authors contributions" section
- Acknowledgments were not included, because there are none
- The "Funding" section was revised
- The supplementary files were cross-referenced and a description added.

With my best regards.

Stefan Kreisel